Author's response to reviews

Title: Effective population management practices in diabetes care - an observational study

Authors:

Anne Frølich (anne.frolich@dadlnet.dk)
Jim Bellows (jim.bellows@kp.org)
Bo Friis Nielsen (bfn@imm.dtu.dk)
Per Bruun Brockhoff (pbb@imm.dtu.dk)
Martin Hefford (loafer@clear.net.nz)

Version: 5 Date: 20 September 2010

Author's response to reviews:

Monday, 20 September 2010
Melissa Norton, MD
Editor-in-Chief
BMC Health Services Research

Dr. Norton:
Please find our answer to reviewers’ comments to the paper entitled “Effective population management practices in diabetes care – an observational study”. MS: 1997682096377743, included in this letter

We have answered the comments raised by the BMC reviewer. We have added that all authors read and approved the final manuscript in the section “Authors contributions”. We have checked the typography and made final corrections.

With all best regards,

Anne Frølich
Senior Consultant, MD, External Lecturer
Department of Integrated Healthcare, Bispebjerg University Hospital
Bispebjerg Bakke 23, stairway 11 B, ground floor
2400 NV København, Denmark
Email:anne.frolich@dadlnet.dk
Phone (direct): +45 35313335